Author's response to reviews

Title: What aspects of quality of primary care predict emergency admission rates? A cross sectional study.

Authors:

Stephen Gunther (stephen.gunther@nhs.net)
Nick Taub (nat2@leicester.ac.uk)
Stephen Rogers (stephen.rogers@northants.nhs.uk)
Richard Baker (rb14@leicester.ac.uk)

Version: 2 Date: 6 June 2012

Author's response to reviews:

Dear Sir,

Please find the revised version of our manuscript. We have in the methods section updated where we obtained the admissions data from (NHS Northamptonshire) with permission. In the practice characteristics section we have added a line to indicate that the GP patient survey id freely available online and finally in final paragraph (above box 2) indicated that we used the practice postcode IMD.

Yours

S Gunther